Author's response to reviews

Title: Uptake and effectiveness of the Children's Fitness Tax Credit in Canada: the rich get richer

Authors:

John C Spence (jc.spence@ualberta.ca)
Nicholas L Holt (nick.holt@ualberta.ca)
Julia K Dutove (jdutove@ualberta.ca)
Valerie Carson (vlcarson@ualberta.ca)

Version: 4 Date: 7 June 2010

Author's response to reviews:

As requested we have done the following modifications to our manuscript:

1. Apart from the fact that we did not have a conclusions section, which we have now added, our headings appear to be consistent with the required structure/format.

2. Line numbers have been removed.

3. Title has been fixed.

4. References fixed (i.e., et al removed and all authors names are included).